Author's response to reviews

Title: Extracellular vesicles: potential applications in cancer diagnosis, prognosis, and epidemiology

Authors:

Mukesh Verma (vermam@mail.nih.gov)
Tram K. Lam (lamt@mail.nih.gov)
Elizabeth Hebert (elizabeth.hebert@nih.gov)
Rao L. Divi (divir@mail.nih.gov)

Version: 3
Date: 24 March 2015

Author's response to reviews:

The article “Extracellular vesicles: potential applications in cancer diagnosis, prognosis, and epidemiology” is commissioned by Dr Magdalena Morawska (Executive Editor) and written based on our conversation during the “2014 Controlling Cancer Summit” conference in London. The focus of the article is on the novel disease diagnostic tools.

On March 4, 2015 Ms Priences Ducot inform me about the acceptance of the manuscript.